Chohan /24-Oct-05
Week 43 starting 24Oct 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  24Oct 05 :  10/24/2005 1:08:58 PM
	4
	A1
	MBAR1275
	PTE Min. NRJ 3KA Diode Test
	4.19 HV

	1
	A2
	MBAL1154
	ICS 5 Traveller Last
	296.43

	6
	B1
	SSS638
	PT 1 IAP @ cold
	1.90

	10
	B2
	SSS162
	Prep 3 CDW1 HV
	295.8

	7
	C1
	MBAL3139
	New Magnet arrived
	294.64

	8
	C2
	MBBL3405
	Prep 5 Cool Down
	95.7

	3
	D1
	SSS292
	PT 8 Fast De-excitation
	1.90

	11
	D2
	SSS604
	PT 12 Warm Up
	295

	2
	E1
	MBBL3404
	PT 12 Warm Up
	280.18

	12
	E2
	CFB_LEAK
	Cryo leak repair
	298.04

	5
	F1
	MBAR2236
	PT 13.2 Resist. Meas.
	300.82

	9
	F2
	MBAL2210
	Prep 5 Cool Down
	280.36


 
   
Coordinator this week: stephane; MTM support: 16-3602

Only 14 complete Magnet Tests cycles  from 11 benches ( E2 still under CFB repair)   10 Dipoles + 3  SSS  + 1 SSSS

 Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 

Situation From Cryogenics:  J-P Lamboy or ACR Rep.

No new magnet tested on E2 since 29 Sept (already 3 weeks !) 

E2 CFB (leak) Repaired  & restart usage ?

Special  Points :
· Special SSS on B1 :  Roy
· Special SSS631  on D2 :   MM Chaconsa or not?

· 2236 HV failure  & investigation ?

AOB 
	

	Magnets under Test from 16-10-2005 to 22-10-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAL2207 
	[image: image1.png]



	


	12-10-2005 
	30 h.
	28 h.
	17 h.
	15 h.
	19 h.
	16-10-2005 
	109 h.
	Stripping
	
	

	2
	TBA1
	MBBR1259 
	[image: image3.png]



	


	16-10-2005 
	37 h.
	34 h.
	13 h.
	12 h.
	23 h.
	21-10-2005 
	119 h.
	Stripping
	
	

	3
	TBA2
	MBAR1265 
	[image: image5.png]



	


	13-10-2005 
	32 h.
	21 h.
	54 h.
	16 h.
	6 h.
	19-10-2005 
	129 h.
	Stripping
	
	

	4
	TBB1
	SSS654 
	[image: image7.png]



	


	15-10-2005 
	74 h.
	29 h.
	10 h.
	16 h.
	14 h.
	20-10-2005 
	143 h.
	Stand By
	
	

	5
	TBB2
	SSS049 
	[image: image9.png]



	


	15-10-2005 
	59 h.
	13 h.
	16 h.
	14 h.
	12 h.
	20-10-2005 
	114 h.
	Stripping
	
	

	6
	TBC1
	MBAL2192 
	[image: image11.png]



	


	16-10-2005 
	36 h.
	24 h.
	34 h.
	28 h.
	0 h.
	21-10-2005 
	122 h.
	Not Found or Thermal cycle
	
	

	7
	TBC1
	MBBR1278 
	[image: image13.png]



	


	11-10-2005 
	43 h.
	22 h.
	28 h.
	15 h.
	16 h.
	16-10-2005 
	124 h.
	Stand By
	
	

	8
	TBC2
	MBAL1070 
	[image: image15.png]



	


	18-10-2005 
	32 h.
	8 h.
	27 h.
	17 h.
	10 h.
	22-10-2005 
	94 h.
	Stripping
	
	

	9
	TBC2
	MBAR2222 
	[image: image17.png]



	


	14-10-2005 
	38 h.
	26 h.
	16 h.
	17 h.
	8 h.
	18-10-2005 
	105 h.
	Stripping
	
	

	10
	TBD1
	SSS163 
	[image: image19.png]



	


	12-10-2005 
	34 h.
	17 h.
	45 h.
	7 h.
	10 h.
	16-10-2005 
	113 h.
	Stripping
	
	

	11
	TBD1
	SSS292 
	[image: image21.png]



	


	16-10-2005 
	49 h.
	26 h.
	60 h.
	-
	-
	22-10-2005 
	135 h.
	Not Found or Thermal cycle
	
	

	12
	TBE1
	MBBL3411 
	[image: image23.png]



	


	15-10-2005 
	32 h.
	22 h.
	18 h.
	17 h.
	10 h.
	19-10-2005 
	99 h.
	Stripping
	
	

	13
	TBF1
	MBAR1268 
	[image: image25.png]



	


	14-10-2005 
	26 h.
	37 h.
	33 h.
	18 h.
	13 h.
	19-10-2005 
	127 h.
	Stripping
	
	

	14
	TBF2
	MBBL2055 
	[image: image27.png]



	


	14-10-2005 
	41 h.
	21 h.
	31 h.
	15 h.
	7 h.
	18-10-2005 
	115 h.
	Stripping
	
	


 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL2207
	10855 A.
(D2_L)
	12572.9 A.
(D1_U)
	12830.7 A.
(D2_U)
	[image: image29.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBR1259
	12640.7 A.
(D2-U)
	12850 A.
[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBAR1265
	11318.8 A.
(D2_U_S01)
	11842.8 A.
(D1_L_S01)
	12461.7 A.
(D2_L)
	12786.2 A.
(D2-L-S13)
	12850 A.
[image: image31.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS654
	5820 A.
[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS049
	12223.2 A.
(QE_12)
	12850 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAL2192
	10734.4 A.
(D2_L)
	11761.9 A.
(D1_L)
	12103.7 A.
(D2-U)
	12276.7 A.
(D1-U)
	12329.9 A.
(D1-L)
	12461.8 A.
(D1_L)
	12568.8 A.
(D2_U)
	11971 A.
(D2_U)
	12742.9 A.
(D1_L)
	[image: image34.png]



	-
	-
	-

	7
	MBBR1278
	11906 A.
(D1_l)
	12578 A.
(D1_L)
	12850 A.
[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAL1070
	11959 A.
(D2_U)
	12850 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBAR2222
	12466.1 A.
(D1-L)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS163
	12244.7 A.
(QE_12)
	0 A.
(No Data)
	12366 A.
(QI-12)
	12710 A.
(QI-12)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-

	11
	SSS292
	11490.2 A.
(QI_12)
	11914.2 A.
(QE_34)
	12162 A.
(QI_12)
	12263 A.
(QE_34)
	12358.3 A.
(QI-34)
	12506 A.
(QI-12)
	12566 A.
(QE-12)
	12678.9 A.
(QE_12)
	12795.4 A.
(QI-12)
	[image: image39.png]



	-
	-
	-

	12
	MBBL3411
	11834.1 A.
(D2-U)
	12727 A.
(D1-L)
	12857 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAR1268
	11715.6 A.
(D2_U)
	12108.8 A.
(D1-L)
	12692.1 A.
(D2_L)
	[image: image41.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBBL2055
	9912 A.
(D2_U)
	11765.6 A.
(D2-U)
	12267.7 A.
(D1-U)
	12684.2 A.
(D2_U)
	12847.7 A.
(D1_L)
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-


[image: image43.png]143

Magnets from Sunday, Oc|

Ref 23 1268 31l 5163

S1

16, 2005 to Saturday, October 22, 2005

[

50




Shows  SSS 163 and 292(both on   D1) and long COLD TESTS Times due to Cohabitation with SpecialSSS on D2
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_18_N
	TBC2


	Mobic 2 displays a message “MTF temporarily unavailable”. Scan of the MBAL1298  performed correctly with MOBIC 1
	MTF
	
	Network not available last week?

	Friday_21_A
	TBB2


	New version of IAP installed in local server did not show the VT readings. It was working properly through TEMA in the control room.
	AB/CO
	IAP
	Giorgio was called. He davised to perform the analysis from control room. He sent the info to Boris Khomenkho  Monday. Problem solved.

	Friday_21_N
	TBD1


	SSS292: Alarm on anticryostat ap 2. The short anticryostat of AP2 (aperture 2) is at 6 C. SSW return wire was trapped inside the anticryostat.
	MTM
	
	Giorgio informed. Salem increased the flow of N2. After investigation Giorgio concluded that ICS did not install the gamelle and MLI

	Friday_21_N
	All


	Unable to cool down or warm up magnets in the gaseous phase because the supervision of the HP compressor is not working (software blocked)
	ACR
	
	11 hours lost.

	Saturday_22_A
	TBB1


	Warm up quench on SSS150 : ALERTS : QH1 (Imax<50, TauIniV>117, TauIniI>120, Rinit>16.5)
	MTM
	AQA
	Giorgio checked the results and asked QH resistance measurements: OK

Magnet cooled down again and warm up quench repeated. When quench analysed from control room it is OK.

	Saturday_22_N
	TBA2


	Cryo loss during HV test on MB1154.

HV test (CDAP) failing at external and internal bus bar vs ground
	MTM
	
	7 hours of stop.

Related?

	Sunday_23_M
	TBF1
	HV test dipole vs ground failed on MB2236.
	MTM
	
	Pierre will ask the mechanics to disconnect the current leads to measure the resistance magnet/ground.
Resistance between GND and current lead after disconnection of the leads is 0.1 ohm. Pierre advised to warm up the magnet slowly. Problem confirmed monday after disconnection. It came from the magnet.

	Monday_24_M
	
	MOBIC1 not working . USB driver error.
	MTM
	
	Rest done by Richard then OK.

	Monday_24_M
	TBD1


	Values of TF measured with SSW are wrong.
	MTM
	
	Richard called. SSW The connector from the return wire was damaged and it was repaired by Fabienne

	Monday_24_A
	TBD1


	XY values not correct because there is a cold point in the short anti-cryostat.
	MTM
	
	Nikolay asked to dry the ttwo aperture with a blower and repeat the measurements fast. XY measurements are suppresses by Stephane. Only transfer function at one current remained.


